Author's response to reviews

Title: The impact and process of a community-led intervention on reducing environmental inequalities related to physical activity and healthy eating - a pilot study.

Authors:

Rachel C Davey (rachel.davey@canberra.edu.au)
Gemma L Hurst (g.l.hurst@staffs.ac.uk)
Graham R Smith (g.r.smith@staffs.ac.uk)
Sarah C Grogan (s.c.grogan@staffs.ac.uk)
Judy Kurth (Judy.Kurth@stoke.nhs.uk)

Version: 2 Date: 26 August 2011

Author's response to reviews:

1) Document, within your manuscript, whether you will be receiving informed consent from your study participants and whether this will be written or verbal.

The statement "written, informed consent will be obtained from study participants" is added to page 16 of the manuscript in response to the above comment. The text is also highlighted in red.

Yours Sincerely,

Gemma Hurst